Author's response to reviews

Title: Primary care provider perceptions of intake transition records and shared care with outpatient cardiac rehabilitation programs

Authors:

Jonathan Yee (jonyee@yorku.ca)
Karen Unsworth (karen.unsworth@lhsc.on.ca)
Neville Suskin (neville.suskin@lhsc.on.ca)
Robert D Reid (breid@ottawaheart.ca)
Veronica Jamnik (ronij@yorku.ca)
Sherry L Grace (sgrace@yorku.ca)

Version: 4 Date: 3 September 2011

Author's response to reviews:

September 3, 2001
Joze Braspennning, Associate Editor
BMC Health Services Research
BMC Series Series Editorial Production Team

RE: REFORMATTED MANUSCRIPT # 1884898907512974

To Whom it may concern:

As requested, we have made additional revisions to the manuscript:
1) Figure one title has been added to the figure.
2) We have revised the references to state all the authors of each article.

Thank you for your correspondance and we hope to hear from you soon.

Sincerely,

Sherry L. Grace
Sherry Grace, PhD
York University, Toronto, ON, Canada
KaHS, Faculty of Health
Bethune 368
4700 Keele St
Toronto, ON M3J 1P3
sgrace@yorku.ca